Author's response to reviews

Title: Retention of the posterior cruciate ligament versus the posterior stabilized design in total knee arthroplasty: a prospective randomized controlled clinical trial

Authors:

Lennard GH Van den Boom (l.vandenboom@home.nl)
Reinoud W Brouwer (r.w.brouwer@mzh.nl)
Inge Van den Akker-Scheek (i.scheek@orth.umcg.nl)
Sjoerd K Bulstra (s.k.bulstra@orth.umcg.nl)
Jos JAM Van Raaij (raayjjam@mzh.nl)

Version: 5 Date: 29 September 2009

Author's response to reviews: see over
Dear Editors,

Thank you for accepting our manuscript for publication in BMC Musculoskeletal Disorders. Hereby, we submit our final revised manuscript.

This letter gives a description of the following and only revision.

In the Abstract section (page 2) we changed desgin to design

We hope our revised manuscript will be suitable for publication within the Study protocol section.

Sincerely,

Lennard GH van den Boom
Reinoud W Brouwer
Inge van den Akker-Scheek
Sjoerd K Bulstra
Jos JAM van Raaij